Author’s response to reviews

Title: Clinical trial on the efficacy of exhaled carbon monoxide measurement in smoking cessation in primary health care.

Authors:

Joana Ripoll (jripoll@ibsalut.caib.es)
Helena Girauta (hgirauta@ibsalut.caib.es)
Maria Ramos (mramos@dgsanita.caib.es)
David Medina (dmedina@ibsalut.caib.es)
Agnès Pastor (apastor@ibsalut.caib.es)
Cristina Alvarez-Ossorio (calvaezossorio@ibsalut.caib.es)
Lucia Gorreto (lgorreto@ibsalut.caib.es)
Maria Esteva (maesteva@ibsalut.caib.es)
Elena García (elenagarcia@ibsalut.caib.es)
Ana Uréndez (amurendez@ibsalut.caib.es)
Ana Buades (ambuades@ibsalut.caib.es)
Elena Torres (elenatorres@ibsalut.caib.es)

Version: 2 Date: 26 March 2012

Author’s response to reviews: see over
Editors in Chief

BMC Public Health
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Palma de Mallorca, 26 March 2012

Dear editor/s,

We are pleased to send you by mail evidence of our approvals:

- Certificate of ethical approval. This study protocol was approved by the Mallorca Primary Care Research Committee and Balearic Islands Clinical Research Ethics Committee (IB 985/08 PI).

- Certificate of external funding. As we declare in the protocol study we have a grant for the study from the Health Research Funds (Fondo de investigación sanitaria (FIS) Carlos III Health Institute (PI09/90841) supported by the Ministry of Health and Consumer Affairs.

- Certificate of Trial Registration: Current Controlled Trials ISRCTN67499921.

We have’nt done any change in the previous version of the protocol study.

This manuscript has not been published or even not sent elsewhere.

We are looking forward your answer,

Your sincerely,

Joana Ripoll
Primary Care Research Unit
C/ Reina Esclaramunda, 9
Palma de Mallorca 07003
jripoll@ibsalut.caib.es